Author’s response to reviews

Title: Cognition after malignant media infarction and decompressive hemicraniectomy

Authors:

Holger Schmidt (H.Schmidt@med.uni-goettingen.de)
Trutz Heinemann (info@lichtenau-ev.de)
Judith Elster (judith.elster@googlemail.com)
Marija Djukic (mdjukic@gwdg.de)
Stefan Harscher (s.harscher@klinikum-weimar.de)
Katja Neubieser (katja.neubieser@gmx.de)
Hilmar Prange (hprange@gwdg.de)
Andreas Kastrup (andreas.kastrup@klinikum-bremen-mitte.de)
Veit Rohde (veit.rohde@med.uni-goettingen.de)

Version: 2 Date: 24 November 2010

Author’s response to reviews:

To
Angelina Ilievska, MSc.
Senior Editorial Assistant
Editorial Office
BioMed Central
Floor 6, 236 Gray’s Inn Road
London,
WC1X 8HL

Dear Mrs. Ilievska,

thank you for your letter. According to your suggestions we have added a phrase in the methods section for the approval of the ethics committee including the approval number. We have also added that all participants have signed and agreed to take part in the study.

We have also extended the financial disclosure section / competing interests section at the end of the manuscript and added a section that describes the respective author's contribution in the study.

Thank you very much for your suggestions,

kind regards,

Holger Schmidt